Author's response to reviews

Title: HGF/c-met/Stat3 signaling during skin tumor cell invasion: indications for a positive feedback loop.

Authors:

Zanobia A Syed (zsyed@lsuhsc.edu)
Weihong Yin (wyin@wfubmc.edu)
Kendall Hughes (khugh1@lsuhsc.edu)
Jennifer N Gill (jrobe5@lsuhsc.edu)
Runhua Shi (rshi@lsuhsc.edu)
John L Clifford (jcliff@lsuhsc.edu)

Version: 4 Date: 11 May 2011

Author's response to reviews:

Cover letter material for final proofing.

Mellissa Norton May 10, 2011
Editor-in-Chief
BMC Cancer

Dear Dr. Norton,

Please find attached our manuscript (MS: 7441088664166834) entitled "HGF/c-met/Stat3 signaling during skin tumor cell invasion: indications for a positive feedback loop.", with the final proofing corrections indicated below.

1. The affiliation listings have been corrected to match journal style.
2. The ‘additional files’ section has been removed.
3. Typographical errors have been corrected.

Once again we greatly appreciate your consideration of our paper.

Sincerely,

John Clifford, Ph.D.
Louisiana State University Health Science Center
School of Medicine-Shreveport
Dept. of Biochemistry and Molecular Biology
1501 Kings Hwy., PO Box 33932
Shreveport, LA 71130
Phone: 318-675-3188
Fax: 318-675-5180
Email: jcliff@lsuhsc.edu